Author's response to reviews

Title: Effects of a training program after surgically treated ankle fracture: A prospective randomised controlled trial

Authors:

Gertrud M Nilsson (gertrud.nilsson@med.lu.se)
Kjell Jonsson (kjellfransjonas@telia.com)
Charlotte Ekdahl (charlotte.ekdahl@med.lu.se)
Magnus Eneroeth (magnus.eneroth@med.lu.se)

Version: 2 Date: 28 May 2009

Author's response to reviews:

The trial has now been registered and allocated the ACTRN: ACTRN12609000327280
Kind regards,
Gertrud Nilsson